Author's response to reviews

Title: Effects of maternal education on diet, anemia and iron deficiency in school-aged children in Korea

Authors:

Hyeon-Jeong Choi (hyunjeong79@hanmail.net)
Hye-Ja Lee (hyejalee@yahoo.co.kr)
Kyung-Hee Park (jypark5587@gmail.com)
Jae-Heon Kang (fmleader@nuri.net)
Jihyun Song (jhsong10@korea.kr)
Han Byul Jang (hanbyeol-black@hanmail.net)
Ju Yeon Park (jypark5587@gmail.com)

Version: 4 Date: 30 September 2011

Author's response to reviews: see over
Dear Editor,

We are submitting our revised version of the article entitled “Effects of maternal education on diet, anemia, and iron deficiency in Korean school-aged children”.

Our manuscript was edited and formatted as your recommendation.

All authors listed have made a substantial contribution to the information or material submitted for publication and read and approved the final manuscript. The manuscript or portions thereof are not under consideration by another journal or electronic publication and have not been previously published.

Please consider it for publication in BMC Public Health.

Sincerely Yours,

Hye-Ja Lee,
Division of Metabolic Diseases, Korea National Institute of Health, Cheongwon-gun, Chungcheongbuk-do, 363-951, South Korea
Tel: 82-43-719-8692, Fax: 82-43-719-8602
E-mail: hyejalee@yahoo.co.kr